Author's response to reviews

Title: Associations between psychological stress, eating, physical activity, sedentary behaviours and body weight among women: a longitudinal study

Authors:

Jennifer Mouchacca (j.mouchacca@deakin.edu.au)
Gavin R Abbott (gavin.abbott@deakin.edu.au)
Kylie Ball (kylie.ball@deakin.edu.au)

Version: 2 Date: 7 November 2012

Author's response to reviews: see over
To the editor,

It is with pleasure that we submit the paper “Associations between psychological stress, eating, physical activity, sedentary behaviours and body weight among women: a longitudinal study” for consideration by BMC Public Health. This paper reports the results of cross-sectional and longitudinal analyses of relationships between perceived stress, weight and weight-related behaviours in a cohort of socioeconomically disadvantaged women.

We confirm that this manuscript is not under consideration elsewhere, nor will it be submitted to another journal whilst under review with BMC Public Health. This study has received governmental research funding and ethical approval, which have been acknowledged in the manuscript. We look forward to your response.

Kind regards,

Kylie Ball (on behalf of all authors)

Professor Kylie Ball (on behalf of all authors)
Personal Chair
NHMRC Senior Research Fellow
Centre for Physical Activity and Nutrition Research
School of Exercise and Nutrition Sciences, Faculty of Health
Deakin University, Burwood